

Using virtual reality to study pedestrian exit choice behaviour during evacuations

Feng, Yan; Duives, Dorine C.; Hoogendoorn, Serge P.

DOI 10.1016/j.ssci.2021.105158

Publication date 2021 Document Version Final published version

Published in Safety Science

Citation (APA)

Feng, Y., Duives, D. C., & Hoogendoorn, S. P. (2021). Using virtual reality to study pedestrian exit choice behaviour during evacuations. *Safety Science*, *137*, Article 105158. https://doi.org/10.1016/j.ssci.2021.105158

Important note

To cite this publication, please use the final published version (if applicable). Please check the document version above.

Copyright

Other than for strictly personal use, it is not permitted to download, forward or distribute the text or part of it, without the consent of the author(s) and/or copyright holder(s), unless the work is under an open content license such as Creative Commons.

Takedown policy

Please contact us and provide details if you believe this document breaches copyrights. We will remove access to the work immediately and investigate your claim.

Contents lists available at ScienceDirect

Safety Science

journal homepage: www.elsevier.com/locate/safety

Using virtual reality to study pedestrian exit choice behaviour during evacuations

Yan Feng^{*}, Dorine C. Duives, Serge P. Hoogendoorn

Department of Transport & Planning, Faculty of Civil Engineering and Geosciences, Delft University of Technology. Stevinweg 1, 2628 CN Delft, the Netherlands

ARTICLE INFO

Keywords:

Evacuation

Validation

Mobile VR

Field experiment

Exit choice behaviour Virtual reality experiment

ABSTRACT

Exit choice is vital to pedestrians' survival during evacuations. This paper presents the results of a VR experiment and a field experiment to study pedestrian exit choice behaviour during evacuations. Primarily, we compared pedestrian exit choice behaviour with a VR experiment and a field experiment to determine the ecological validity of a particular VR simulator (smartphone-based HMD and 360° video) as a research tool to study pedestrian exit choice behaviour. The results showed that the pedestrians' exit choice behaviour during the evacuation is similar in the field experiment and the VR experiment. Furthermore, we investigated whether and to what extent different types of information (i.e., exit signs, directional signs, presence of people) influence pedestrian exit choice during evacuations. The analysis focused on the commonalities and differences in the pedestrians' exit choice behaviour between the scenario without additional information and three scenarios with different types of information. The comparison between scenarios with different types of information illustrated that the presence of other pedestrians and directional signs have a significant influence on the participants' exit choice. Moreover, the results indicated that this VR simulator is applicable to study pedestrian exit choice behaviour during evacuations.

1. Introduction

When evacuating a building, pedestrians usually face multiple exits and need to choose which of the exits to use (Heliövaara et al., 2012). Choosing the right exit is crucial for survival; hence it is important to understand such so-called pedestrian exit choice behaviour (Kobes et al., 2010b). Thus, for many disciplines, such as architecture, fire safety, or civil engineering, it is vital to have a thorough understanding of pedestrian exit choice behaviour to ensure pedestrian safety and create safe building designs.

During building evacuations, the information provided in the environment is a key aspect to provide evacuees clues to find an exit. A number of prior studies have found that information provided by the signage features or other pedestrian's influence pedestrian exit choice behaviour. For instance, visibility of evacuation exit signs (Haghani and Sarvi, 2016a; Kobes et al., 2010b; Wong and Lo, 2007), signage about exits' direction (Bode et al., 2014; Ronchi et al., 2016; Vilar et al., 2013), and the indirect information provided by the presence of other people (Haghani and Sarvi, 2016a; Helbing et al., 2000; Kinateder and Warren, 2016; Lovreglio et al., 2016; Moussaïd et al., 2016). Therefore, it is also crucial to understand to what extent different types of information influence pedestrian exit choice behaviour during evacuations.

In recent years, different experimental studies have been performed to study pedestrian exit choice behaviour during evacuations. These corresponding experimental methods predominantly included observations of real-life evacuation situations (e.g., Galea et al., 2017a, 2017b; Kobes et al., 2010b; Shields and Boyce, 2000), controlled experiments under conditions of stress (e.g., Fang et al., 2010; Fridolf et al., 2013; Zhu & Shi, 2016) and surveys (e.g., Duives and Mahmassani, 2012; Haghani and Sarvi, 2016b; Lovreglio et al., 2014). Although these experimental studies have provided valuable information regarding the understanding of pedestrian exit choice behaviour in emergency situations, there are stringent limits regarding the insights one can derive by means of field observations or controlled experiments. For instance, it is difficult to control external variables in field observations, and evacuation drills cannot be completely realistic because of ethical and financial constraints. In controlled experiments, there are ethical and financial constraints to create real and stressful situations. Meanwhile, it is hard to provide participants with a strong sense of presence to make them fully participate and keep focused on the task which constrains the

* Corresponding author. E-mail addresses: y.feng@tudelft.nl (Y. Feng), d.c.duives@tudelft.nl (D.C. Duives), s.p.hoogendoorn@tudelft.nl (S.P. Hoogendoorn).

https://doi.org/10.1016/j.ssci.2021.105158

Received 6 August 2020; Received in revised form 6 November 2020; Accepted 31 December 2020 Available online 25 January 2021 0925-7535/© 2021 The Author(s). Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).







generation of realistic evacuation behaviour. Besides that, the answers from participants in surveys may deviate from their actual actions in real situations, especially the case when respondents are required to answer questions regarding unfamiliar situations or are required to recall past experiences.

To overcome these constraints, Virtual Reality (VR) has gained increasing attention and popularity for investigating pedestrian behaviour during evacuations. With VR experiments, it is possible to study the behaviour of participants during emergencies safely, while maintaining experimental control to analyse the influence of different factors on pedestrian behaviour more precisely. A number of prior VR studies have shown promising results regarding the derivation of pedestrian behaviour during evacuations (e.g., Duarte et al., 2014; Kinateder et al., 2019; Kinateder and Warren, 2016; Kobes et al., 2010a; Tang et al., 2009). The application of VR technologies is, therefore expected to be a valuable complement to the current experimental method toolbox to study pedestrian evacuation behaviour.

However, before VR can be adopted as a methodology for evacuation research, its ecological validity must be established (Kinateder and Warren, 2016). Although some aspects of pedestrian behaviour during evacuations have been investigated using VR, there are very few studies that validate VR research methods (e.g., Kinateder and Warren, 2016; Kobes et al., 2010a). Thus, in order to develop VR method as a valid research approach to study pedestrian behaviour, empirical evidence is needed that compares pedestrian behavioural results collected in a virtual world with behavioural results collected in the real world.

This study has two objectives, namely (1) to validate whether Virtual Reality combing smartphone-based HMD and 360° video can be used to measure pedestrian exit choice behaviour during an evacuation, and (2) to identify the impact of information on pedestrian exit choice during evacuations. To validate this VR, this study compares pedestrian exit choice behaviour during a real-life evacuation drill and a VR experiment that covers exactly the same situation. Afterwards, the VR simulator is used to determine to what extent different types of information (i.e., visibility of exit signs, directional information and the presence of other people) influence pedestrian exit choice behaviour.

The outline of the paper is as follows. Section 2 provides an overview of the state of the art of using different experimental methods to study pedestrian exit choice behaviour during evacuations. Section 3 presents the research methodology for the field experiment and the VR experiments. Section 4 focuses on validation results obtained from field observation and VR experiments. Section 5 presents the results featuring the impact of information on pedestrian exit behaviour during evacuations. Section 6 discusses the results of the validation study and the effect of different types of information on pedestrian exit choice behaviour. This paper ends with conclusions, limitations and suggestions for future research.

2. Background

Up to this moment, predominantly three experimental methods have been used to study pedestrian exit choice behaviour during evacuations, including traditional methods such as field observations in real life, traditional controlled experiments, VR experiments and surveys. This section provides a brief overview of the work featuring experimental methods that make use of revealed behavioural data to study pedestrian choice behaviour during evacuations with a focus of field observations, controlled experiments and VR experiments. Consequently, this section reviews research pertaining to field observations, traditional controlled experiments and VR experiments that have been used to specifically study pedestrian exit choice behaviour.

2.1. Using field observations to study pedestrian exit choice behaviour during evacuations

Field observations are the traditional method to study pedestrian

(exit choice) behaviour in real life. The intention of field observations is to study pedestrian behaviour as unobtrusively as possible (Feng et al., 2021). Usually, a space with multiple exits is chosen as the experimental area and digital devices (e.g., camera) are used to record the evacuation process. For instance, Proulx (1995), Shields and Boyce (2000) and Galea et al. (2017a, 2017b) used video recordings to study pedestrian exit choice behaviour in apartment buildings, retail stores and a theatre. More recently, Imanishi and Sano (2019) and Rahouti et al. (2020) analysed evacuees' movements during evacuation drills in a theatre and a hospital respectively, featuring pedestrian route and exit choice. These studies illustrate that field observations can be a valuable experimental method to understand exit choice behaviour in the real world because pedestrians are more likely to behave naturally. Nevertheless, the variables of field experiments are difficult to control and evacuation drills cannot be completely realistic due to ethical and financial constraints. Therefore, it would be difficult for researchers to set up field observations to investigate how external variables (e.g., signage) influence pedestrian exit choice behaviour during evacuations.

2.2. Using traditional controlled experiments to study pedestrian exit choice behaviour during evacuations

Compared to field observations, in traditional controlled experiments (i.e., laboratory experiments and field experiments), variables can be controlled and the effect of each separate factor can be observed and analysed. Traditional controlled experiments have been widely used to study pedestrian behaviour under stressed conditions (e.g., enforce participants to hurry). One group of studies focused on the analysis of pedestrian exit choice under various experimental settings. For instance, Fang et al. (2010), Heliövaara et al. (2012), Zhu and Shi (2016) studied evacuees' exit selection in a hall, a corridor and a classroom respectively. Kobes et al. (2010b) observed pedestrian behaviour during an unannounced evacuation drill in a hotel and investigated the influence of exit signs on pedestrian exit behaviour. Fridolf et al. (2013) and Ronchi et al. (2018) investigated pedestrian movement and exit choice in tunnel evacuation experiments with smoke. Another group of studies focused on the study of the influence of signage on pedestrian exit choice behaviour during evacuations. For instance, Wong and Lo (2007) tested the visibility of different exit signs during emergencies and Galea et al. (2017a, 2017b) studied the effect of dynamic signage on pedestrian exit choice behaviour. Guo et al. (2012) and Zhu and Shi (2016) performed classroom evacuation experiments to contrast route and exit choice behaviour under varying visibility conditions, occupant distributions and alarm information. More recently, Cao et al. (2018) investigated exit behaviour of pedestrians during evacuation under good and limited visibility in a supermarket. These studies illustrated that controlled experiment is also a useful approach to improve our understanding of pedestrian evacuation behaviour and the effect of external variables (e. g., information and infrastructure) on the choice behaviour of pedestrian. Nevertheless, participants have prior knowledge of the experiment and might not act naturally during the experiment. Furthermore, it is often costly to set up and perform controlled experiments.

2.3. Using VR experiments to study pedestrian exit choice behaviour during evacuations

To (partially) overcome the constraints of traditional experiment methods, VR is becoming a promising experimental approach that can potentially be used to study evacuation choice behaviour. With VR, it is possible to safely study the behaviour of participants during emergencies in immersive virtual environments, without being exposed to health risks (e.g., fire and smoke). Meanwhile, VR allows researchers to easily build and change the virtual scenes, thus a number of possible factors that potentially influence pedestrian behaviour can be studied with high experimental control (Feng et al., 2021). The development and the increasing availability of VR hardware (e.g., HTC Vive, Oculus Rift, Samsung Gear VR) and software (e.g., Unreal Engine 4, Unity) provide different techniques and applications for VR to study pedestrian behaviour. Generally, two types of VR simulators exist, namely non-immersive VR and immersive VR.

In non-immersive VR, the virtual environment is displayed on a device, for instance, desktop monitor, and individuals interact with the environment via a device that controls the simulator (e.g., mouse, keyboard, joystick). For instance, Tang et al. (2009) created a VR emergency escape game to determine if and how various emergency signs help pedestrian behaviours in way-finding. Bode et al. (2014) tested the influence of directional signs on pedestrian exit choice. Silva et al. (2013) used a serious game to investigate pedestrian evacuation time and exit choice in a hospital while the game system provided information regarding the evacuation. Bode et al. (2015) and Fang et al. (2020), moreover, investigated the impact of presence of other people on exit and route choice behaviour using a 2D simulated virtual environment.

Another type of VR simulator is the immersive simulator. In immersion VR simulators, the virtual world is presented in a way that the virtual environment surrounds the participants. The participant interacts with the virtual environment through specialist simulator control devices (e.g., joystick, gloves) and motion tracking hardware (e.g., eye, head and motion tracking devices).

The first commonly adopted immersive simulator is the Cave Automatic Virtual Environment (CAVE), which displays the environment on huge screen monitors or multiple television projection systems simultaneously (Mujber et al., 2004). For instance, Kobes et al. (2010a) studied pedestrian exit and route choice behaviour in a hotel. Kinateder et al. (2014a, 2014b) used a 3D CAVE system to investigate the effect of social influence on pedestrian route and exit choice during evacuation from a tunnel emergency. Ronchi et al. (2016) investigated the effect of different lighting conditions during a road tunnel evacuation. Another most often used immersive simulator is HMD (head-mounted display). One major research theme using HMD simulators focused on studying the influence of environmental variables on pedestrian exit choice behaviour. Amongst other studies, Duarte et al. (2014) examined how dynamic features in exit signs affect pedestrian exit behaviour during an emergency egress using HMD device and a joystick. Moreover, Cosma et al. (2016) studied the influence of varying lighting conditions on pedestrian behaviour in a virtual tunnel evacuation and Tucker et al. (2018) tested the influence of hazard level and obstacle information on evacuees' anxiety levels and exit choice in a fire evacuation. Kinateder et al. (2019) studied the impact of coloured signs on pedestrian exit choice behaviour. Cao et al. (2019) and Zhu et al. (2020) studied the impact of visual cues on pedestrian wayfinding behaviour, where the first studied the impact of virtual fire and the latter studied the impact of visual access. In a more recent study, Zhao et al. (2020) analysed the effects of different crowd management strategies (e.g., remove fences and/or police cordons) on crowd behaviour for the 2010 Love Parade disaster. Another theme focused on the social influence on pedestrian exit choice behaviour. For instance, Kinateder and Warren (2016) studied the effect of social influence on pedestrian exit choice and Kinateder et al. (2018) used a wireless head-mounted display to investigate the influence of exit familiarity and neighbour's behaviour on exit choice in a virtual ambulatory museum. More recently, Lin et al. (2020) examined the influence of crowd flow on pedestrian route and exit choice behaviour during evacuations.

These studies illustrate that, although some aspects of pedestrian behaviours during evacuation have been investigated, there are few studies that validated VR research methods. That is, very few studies have compared pedestrian behaviour in VR and real life. For instance, Kobes et al. (2010a) compared pedestrian evacuation behaviour in a real-life hotel and a virtual hotel. Subsequently, Kinateder and Warren (2016) compared pedestrian exit choice behaviour in a real and virtual environment to establish the ecological validity of VR. More recently, Li et al. (2019) investigated pedestrian route choice behaviour around obstacles in a field experiment and an identical 2D virtual environment. Given the little evidence pertaining to the validity of VR simulators to study pedestrian choice behaviour, there is a need for studies that determine whether VR is a valid and reliable technique to study pedestrian choice behaviour during evacuations.

Moreover, although above-mentioned PC-based and console-based HMD devices can provide participants with a rich immersive experience, the cost of a comprehensive immersive system can be relatively high. A potential low-cost solution is smartphone-based HMD which is the combination of an HMD device and a smartphone (Krevelen and Poelman, 2010). To the author's knowledge, only one study has used smartphone-based HMD to investigate pedestrian evacuation behaviour (Ronchi et al., 2019). Their results showed the consistency between a low-cost HMD and a CAVE system, which is a first attempt at establishing the validity of this low-cost immersive solution. However, the feature of different VR devices (e.g., interaction function and immersion) and costs vary, thus more smartphone-based HMD studies are needed in order to help researchers understand the validity and the potential of using cost-efficient smartphone-based HMD to investigate pedestrian evacuation choice behaviour.

3. Experimental method

To achieve the first objective of this study, two experiments were carried out to study the exit choice behaviour of pedestrians during evacuation when they face either a real-life environment or a virtual environment. To achieve the second objective, we compared the commonalities and differences in exit choice between scenarios with different types of information in the VR experiments. Underneath these two experiments are further specified. First, the setup of the field experiment is described in Section 3.1. Accordingly, the setup of the VR experiment is detailed in Section 3.2.

3.1. Field evacuation experiment

The video recordings were captured during an unannounced evacuation drill conducted on November 23rd, 2017 as part of the safety training programme of the emergency response services of the faculty of Architecture of the Delft University of Technology. The subsequent analysis mentioned in this paper was conducted as part of this training exercise to evaluate the student's and safety officer's response to the alarm. As such, no Human Research Ethics Committee (HREC) examination was required for the evacuation drill. The subsequent usage of the video for the VR experiment was assessed by the HREC as part of the evaluation of the VR experiment protocols. See Section 3.2 for the details pertaining to this assessment. Underneath the experimental layout, experiment setup and procedures of this field evacuation experiment are further elaborated upon.

3.1.1. Experiment area layout

A workshop area with multiple exits was chosen as the location of the field experiment. This area was used by students to work on their graduation assignments for the study of architecture. The size of the workshop area was approximately 50×30 m. Both the workshop area and exits were located on the ground floor of the Architecture Faculty (Fig. 1). The workshop area had eight exits. Exits A1 and A2 were located at the front middle of the area and exits B, C&D, E&F and G were located at the four corners of the area. Exit signs were equipped above the exits with a green-coloured background. All eight exits are emergency exits.

At the time of the evacuation drill, twenty-four students were present. Students were standing/sitting next to the tables doing their assignments individually, their locations are identified by the black circles in Fig. 1. While working in this space, students could see all exits from their current location. We do not know the used entrance of each student, but most students generally walk in via Exit A and C, as these exits



are the major exits of the workshop space.

3.1.2. Experiment setup

The evacuation behaviour of the students working in the workshop was observed and recorded for two purposes. First and foremost, 360° videos were needed as the material to create the evacuation scenario for VR experiment. Second, the students' behaviour provided the benchmark for the comparison between the 'real' choice behaviour and the choice behaviour of the participants in the VR experiment.

A combination of normal cameras and 360 cameras was used, namely two 360° cameras (i.e., Nikon KeyMission 360 Camera and Kodak Pixpro SP360 4 K Camera) and three normal cameras (i.e., Car-CamDoo camera). Their positions in the workshop space are identified by icons in Fig. 1. Two of the three 'normal' cameras were placed at the second level of the workshop space, and the third one was placed on a balcony overseeing the workshop space (Fig. 2). These normal cameras had a higher and wider view, which ensured capturing the evacuation process of each individual. In order to identify every person clearly, the two 360 cameras were placed at the height of 1.8 m above the ground to capture the overall movement of the participants from an aerial view (Fig. 3). The setup of the field experiment ensured the behaviour of all students could easily be observed throughout the whole evacuation process without the observation being invaded or disturbing their



Fig. 3. A screenshot of 360° view of the workshop space by the Nikon 360 Camera.

natural behaviour.

All behaviours of the pedestrians (e.g., pre-evacuation behaviour, exit choice) from 5 min before until just after the end of the evacuation drill was recorded and afterwards transcribed. No human intervention within the workshop space was required to activate the cameras.

3.1.3. Experiment procedure of the evacuation drill

The full-scale unannounced evacuation drill of the Architecture



Fig. 2. Screenshots of the view of the workshop space by the normal cameras which were located (a) at the corner and (b) the balcony.

Y. Feng et al.

Faculty building took place on 23rd November 2017, at 11:13 a.m. In the morning of the evacuation drill, researchers set up the experiment. Questions of students and staff members pertaining to the research installation were answered by stating that we were testing a new video tool.

Before the evacuation drill started, students were performing their assignments at the tables in the workshop space as usual. They were not informed about the pending evacuation drill. At the beginning of the evacuation drill, the evacuation alarm went off, followed by a voice message. In line with the normal operating procedure of during an evacuation training at the Delft University of Technology, voice instructions were broadcasted throughout the entire building repeatedly, asking all people to evacuate the building. The alarm message consisted of a female voice that repeated the following statement: "Attention, please leave the building using the emergency exits as indicated. Do not use the elevators.". After three minutes, two members of the security staff walked into the workshop area to ensure that students actually left the space.

3.2. Evacuation experiment in the virtual environment

The VR experiment was conducted during the International Festival of Technology on 6th, 7th, 8th June 2018 in Delft, the Netherlands. The behavioural data that was collected during these three days was used for the validation study as well as the study into the impact of different types of information. The VR experiment was approved by the HREC of the Delft University of Technology (Reference ID No. 422).

3.2.1. Experimental design

A single-factor between-subjects experimental design was used for this study to reduce the learning effects due to repetitive exposure. In order to determine the effect of various types of information on pedestrian exit choice behaviour, four different scenarios were carried out: one scenario to validate the tool and three experimental scenarios (i.e., no additional information, exit sign scenario, direction scenario). Each scenario used the 360° video-recording of the abovementioned full-scale unannounced evacuation drill as the benchmark. In essence, participants were immersed in a 360° visual environment, which was the video-recording from the Nikon KeyMission 360 Camera. Participants started the experiment at the static vantage point of the video recording (Nikon 360 camera, see Fig. 1), which is the same location for participants to perceive the virtual environment and locations of eight exits during the experiment. At the start of the experiment, all participants were facing the tables on the right, as indicated by the white arrow in Fig. 3. The white arrow points to the direction where participants looked at when they were first immersed in the virtual environment. Participants had a 360° view of the environment via rotating their head in real life. That is, participants could not move through the virtual environment. Instead, they could only move their field of view by rotating their head. The installation height of the camera (1.80 m) ensured that participants have a realistic vantage point of the entire workshop area. An alarm sound of the evacuation drill video (Nikon 360 camera) was added to all scenarios in the VR experiment in order to ensure participants received exactly the same information as in the field experiment. As at least two exits are clearly visible from the vantage point of the participants at the start of the VR experiment, we do not expect the notion that there were several exits to induce a bias in the participant's responses.

Besides the four scenarios, a general familiarisation scenario was developed to allow participants to become familiar with the way to observe the virtual environment and the sensation of VR. The familiarisation scenario entailed a 360° video of the workshop space, and it excluded any evacuation clues, evacuation alarm, or any pedestrians. In order to help participants to describe their exit choice, "Left" and "Right" signs were added on the ground of all scenarios (Fig. 4). The participants did not have a visible presence in the virtual environment (i. e., a participant cannot see their own body).



Fig. 4. Screenshots of the Validation scenario.

3.2.1.1. Validation scenario. In order to test the ecological validity, the Validation scenario presents participants with an identical evacuation scenario as the real-life evacuation drill. Thus, the students present during the evacuation drill were also present in the 360° videos, as shown in Fig. 4.

3.2.1.2. Experimental scenario. Next to the Validation scenario, three other scenarios were created to investigate the effect of different types of information on pedestrian exit choice behaviour. The three scenarios featured one without additional information, one with increased visibility of exit signs, and one with additional directional information. The video recording of the validation experiment was used as a benchmark to create these scenarios, which allows us to directly compare the participants' exit choice in three scenarios. All pedestrians appeared in the video recording were removed in the experimental scenarios. For a depiction of the visual changes to the scenario, see Fig. 5. The changes per scenario are:

- *No information scenario*: No additional information was added to this scenario (Fig. 5a).
- *Exit sign scenario*: the concept of increasing visibility was incorporated by adding eight emergency exit signs with larger dimensions than the original signs to the environment. The signs were entirely text-based, containing a green background and white text (Fig. 5b). They were added on top of the original exit signs at the same location.
- *Direction scenario*: four white arrows were added on the floor in front of the view of participants, which point to four directions directly connected exits (i.e., exit B, C, D, E, F and G) and outside of the building (Fig. 5c). The arrows were intended to inform the participants about the location of exits which were less easily observed (compare to exit A1 & A2).

3.2.2. Apparatus

This particular VR experiment does not require highly complex and expensive tools because the collected data was focused on pedestrian exit choice behaviour. Thus, a smartphone-based head-mounted display (HMD) device was deemed sufficient to capture the exit choice of participants. The smartphone-powered HMD device comprises an iPhone X and a VR Pro Virtual Reality Glasses. Participants were immersed in the virtual environment via the HMD, which has an approximate 90° horizontal and 110° vertical field of view. Fig. 6 shows the front view and the top view of the HMD device. The screen was 14 cm length and provided a resolution of 1125 \times 2436 pixels for 3D effects. It has a refresh rate of 90 Hz.

3.2.3. Data collection

The collected data through the VR experiment was mainly two-fold: (1) exit choice behaviour and (2) participants' experience regarding the VR experiment. As the objective of this study was related to exit choice behaviour, the behavioural outcome variables of the VR experiment were the number of exits a participant had identified during the



(a) No information scenario

(b) Exit sign scenario

Fig. 5. Screenshots of three experimental scenarios.



Fig. 6. The front view (a) and the top view (b) of the head-mounted display was used during the VR experiment.

experiment and the actual exit choice of each participant.

In order to obtain participants' exit choice and experiences of participants regarding the VR experiment, participants were asked to complete a questionnaire immediately after the experiment. The questionnaire contained four sections, namely (1) Exit choice behaviour, (2) Simulator Sickness Questionnaire (SSQ), (3) System Usability Scale (SUS) and (4) Presence Questionnaire (PQ).

The first section of the questionnaire recorded participants' exit choice behaviour, in particular, the number of identified exits, final exit choice, and the reason for choosing a certain exit choice. The identified exit is defined as the number of exits participants were able to observe during the experiment. The final exit choice is defined as the final exit participants chose to use at the end of the experiment. The participants were also asked to draw the identified exits immediately after finishing the task on a paper, which is a simplified version of Fig. 1. In addition, some personal details were collected, such as age, gender, familiarity with the building of the Architecture Faculty and previous experience with VR.

The second section featured the Simulator Sickness Questionnaire (Kennedy et al., 1993), which determined if participants experience sickness, such as nausea, oculomotor discomfort, and disorientation, throughout the experiment. Participants reported on a 4-point Likert scale from 0 (no) to 3 (severe) about how much each symptom affected them.

The third section was the Presence Questionnaire (Witmer and Singer, 1998), which assessed the user's feeling of presence in the virtual environment. It includes five factors, namely involvement, immersion, visual fidelity, interface quality and sound effect which influence user presence in a virtual environment. The items were rated on a 7-point scale. It contains questions such as, "How involved were you in the virtual environment experience?", "How completely were all of your senses engaged?", and "How much did your experiences in the virtual environment seem consistent with your real-world experiences?".

The fourth section was the System Usability Scale (Brooke, 1996), which assessed the usability of the applied VR system as a pedestrian simulator. The system usability scale consists of 10 items, which were rated on a 5-point Likert scale (i.e., 1 = strongly disagree, 5 = strongly agree). It includes questions, for instance, "I thought the system was easy to use.", "I felt very confident using the System." and "I thought there

was too much inconsistency in this system.".

3.2.4. Recruitment of participants

A random sampling approach was applied to recruit participants. Firstly, the festival organiser posted information regarding the experiment on their website. Besides that, a news item featuring the opportunity to try VR was spread via social media: LinkedIn, Facebook and WhatsApp. In addition, the experiment was promoted through the traditional media outlets of the university (i.e., posters and digital news feeds). Finally, participants were also recruited during the festival by the researcher. In total, 94 participants volunteered to take part in the VR experiment.

The information provided at the recruitment stage included a description of the experiment and formal details on the location and the dates of the experiment. Please note, the specific features of the VR experiment were not communicated to the potential participants, e.g., the experiment features an evacuation drill, the experimental building (the Architecture Faculty), or the conditions of the evacuation drill.

3.2.5. Experiment procedure

Each participant was randomly assigned to one of four scenarios. The procedure of the VR experiment included four parts, being: (1) introduction, (2) familiarisation, (3) actual evacuation drill, and (4) questionnaire. Underneath the procedure is further detailed.

Introduction: When participants showed interest and indicated that they would like to join the experiment, they were required to read the instructions of the study and were informed of the purpose of the experiment (i.e., to investigate the use of virtual reality for pedestrian behaviour), and their assignment (i.e., to determine which exit to choose). No further information was provided about the scenario that they would experience (i.e., that they would be confronted with an emergency) since this might have led to biases in the participants' response.

VR familiarisation: Accordingly, the participant was instrumented with the head-mounted display. Afterwards, two videos were loaded. Participants firstly explored a familiarisation virtual reality video for 30 s. If participants got sick during this period, they were allowed to have a break, and after the break they could decide whether to continue or quit the experiment.

Evacuation: After the first video, the screen faded to black, and after an interval of 7 s the trial video started. At the start of the evacuation experiment, participants were asked to choose an exit within 3 min. This time limit is based on the time that all pedestrians in field observation required to exit from the building. As such, we do not expect that setting a time limit in the experiment will produce a bias in the choice behaviour due to the time pressure. Participants who got sick during this period were taken out of the experiment because they would already have experienced the evacuation situation. Furthermore, each participant only experienced one scenario, because prior knowledge about the scenario and repeated exposure to the environment is found to influence their exit choice behaviour (Lin et al., 2019; Vilar et al., 2013).

Questionnaire: After the evacuation experiment was finished, participants were asked to remove the head-mounted display and fill in the questionnaire. After finishing the questionnaire, participants were thanked and some sweets were provided (Fig. 7).

4. Results regarding the comparison of exit choice in real life and $\ensuremath{\mathsf{VR}}$

In this section, the results of the field experiment and the Validation scenario from the VR experiment are detailed. First, the characteristic of the sampling population is presented. Afterwards, the exit choice from both experiments is provided. Next to that, the analysis of ecological validity is presented.

4.1. Characteristic of the sampling population

The characteristic of the pedestrians from the field experiment were derived from the video recordings. At the start time of the alarm, 24 individuals were inside of the workshop space performing their assignments, of which 13 were female (54%) and 11 male (46%) students of the Architecture Faculty. All individuals were bachelor or master graduation students, so the age distribution of the participants was between 21 and 25 years old. Given that the workshop area is a room in which students perform their assignments, the pedestrians involved in the real-life evacuation drill were relatively familiar with the structure of the building.

In the Validation scenario from the VR experiment, 27 individuals participated. One participant did not make the exit choice before the experiment ended, thus the results of the Validation scenario discussed underneath are based on 26 participants including 10 females (48%) and 16 males (52%). The ratio of the distribution of gender of the participants between the field experiment and VR experiment was not significantly different (p = 0.204).



Fig. 7. Participants were (a) experiencing the VR experiment and (b) filling in the questionnaire.

Moreover, the distribution of the participants's age in the Validation scenario is slightly different than the field experiment, 16 individuals were in the age range of 18–25 years old and 10 individuals were in the age range of 25–45 years old. Furthermore, of the individuals that participated in the Validation scenario, only 2 participants claimed they are working or studying at the faculty, 12 participants visited there once, and the remaining 12 participants indicated that they had never been in the building before. Results from the Pearson chi-square test demonstrated that there was no significant influence of age (p = 0.310), and familiarity (p = 0.123) on exit choice in the Validation scenario.

4.2. Exit choice behaviour in real-life and VR

Table 1 shows the exit choices obtained in the real-life evacuation drill and the VR experiment. During the real-life evacuation drill, most pedestrians either chose exit A1 or C. The other exits were not being chosen. The final exit choice of each individual was visualized in Fig. 8. It shows that students did not always choose the nearest exit, which implies that distance is not the only factor that influences their exit choice behaviour. To gain insights into the reasons, we analysed the video recordings in detail. In most cases, before starting their decisive evacuation movements to an exit, students decided to undertake a variety of activities, such as packing belongings, searching for others, investigating cues, and seeking confirmation from others. In total, 13 pedestrians showed waiting behaviour and moved to an exit with one or two other pedestrians together. That is, these particular individuals were stalling while waiting, facing and talking to other individuals in the space, before walking out together with the individual they were communicating with. Thus, the students performed so-called herding behaviour relatively frequent (Helbing et al., 2000), where students tend to follow friends or classmates to the exit.

In the Validation scenario from the VR experiment, similarly, most participants chose exit A1 and C. Some participants also chose exit E. No other exits were chosen. The results of the questionnaire illustrated that herding behaviour was a major source of influence on participants' exit choice behaviour. That is, 8 participants claimed they chose the exit because they saw other people went towards the exit. Besides that, the distance and visibility of the exits were also taken into consideration by the participants. For instance, 7 participants chose the exit according to the nearest distance, and 7 participants chose the exit because of good visibility of the exit and their directness towards outside. Furthermore, the familiarity of the participants was also identified as an influencing factor, 2 participants knew the building so they chose it according to their habit.

4.3. Ecological validity analysis

To determine whether this VR simulator (i.e., the combination of smartphone-based HMD and 360° video) can be used to measure pedestrian behaviour, a justification for ecological validity is needed. In other words, in order to use the VR method for future experiments, it is important to determine whether participants behaved similarly in the virtual environment as they would in the real world (Deb et al., 2017). In this study, because exit choice is a categorical variable, the ecological validity of the results generated from the VR experiment was tested using the Pearson chi-square test. The null hypothesis is formulated as

The exit choice in the field experiment and	the Validation scenario.
---	--------------------------

Exits	A1	С	Е	Total
Field experiment	15 (62.50%)	9 (37.50%)	0 (0%)	24
Validation scenario	13 (50.00%)	10 (38.46%)	3 (11.54%)	26
Total	28	19	3	50



Fig. 8. The exit choice of each individual in the field experiment.

follows: the exit choice behaviour during evacuation does not depend on the experimental method. In order to meet the requirement of the chisquare test for its appropriate use (i.e., no more than 20% of the exits should be chosen less than 5 times), exit A1, B, G are combined into one category. That is because exit A1, B, G were all located at the right side of the workshop area, of which only exit A1 was chosen in both field experiment and Validation scenario. Thus, it results in 2 degrees of freedom for the chi-square distribution.

The Pearson chi-square test with a $\chi 2$ value of 3.12 showed that the probability value is 0.21 at a significance level of 95%, thus we cannot reject the null hypothesis. This indicated that the differences between the two experimental methods did not have a significant influence on the exit choice behaviour of the participants.

Another interesting finding is that in both experiments, the pedestrians' exit choice was asymmetrical. Although eight exits were available and all the students or participants could see all exits from their position, exit A1 and exit C were used more often than the other six exits. Moreover, the percentage of the participants that chose exit A1 was higher than the percentage that chose exit C in both experiments. Although we do not know the exact reason why pedestrians chose a particular exit in the real-life evacuation drill, the video recording shows that in total 13 pedestrians waited, talked with other students and moved towards exits A1 and C as two-person or three-person groups (Table 2). In the Validation scenario, 8 participants indicated that they saw other people move towards exit A1 and C. Thus, it could be argued that pedestrians in both the field experiment and the Validation scenario followed the choice behaviour of other pedestrians, which indicates that the VR method can be used to measure pedestrian exit choice behaviour in a qualitative way.

Although the quantitative and qualitative results of exit choice are very similar for both experiments, there are some differences in the participants' exit choice behaviour that cannot be ignored. In the reallife observation, besides exit A1 and exit C, the other exits were not used at all. In comparison, 3 participants in the Validation scenario chose exit E, either because its directness towards outside or because they were very familiar with the environment. Based on the data

Table 2	
---------	--

The reasons for choosing a certain exit in the two experiments.

Exits	Reasons	A1	С	Е	Total
Field experiment	Following others	10	3	0	13
	(Unknown)	5	6	0	11
Total		15	9	0	24
Validation scenario	Following others	5	2	1	8
	Distance to exits	3	4	0	7
	Visibility of exits	2	1	0	3
	Familiarity of exits	1	0	1	2
	Path to exits is clear	1	0	0	1
	Direct exits to outside	0	3	1	4
	Others	1	0	0	1
Total		13	10	3	26

collected during the two experiments, we cannot determine why the pedestrians in real life did not choose these exits. A possible explanation of the differences is that exits A1 and C were closest to the bicycle parking area, which is the most dominant travel mode of the students. Thus, we still expect familiarity might be an influencing factor, although this was not ratified by the results of this study.

5. Results regarding the effect of information on exit choice behaviour

The previous section indicates that the VR experiment generated valid results pertaining to the exit choice in this particular VR setting. This section presents the results of using this VR experiment to investigate the impact of information on exit choice behaviour. First, an analysis of the participants' characteristics is presented in Section 5.1. Secondly, an analysis of questionnaire data is presented in Section 5.2. Afterwards, the exit choice behaviour over four scenarios is analysed in Section 5.3.

5.1. Characterisation of participant population

In total, 95 participants took part in the VR experiment, of which only 94 were accounted for the data analysis because one individual failed to make the exit choice before the experiment ended (as mentioned in Section 3.1). Table 3 presents the descriptive statistics of participants and their distribution over four scenarios. No significant differences were found according to the Fisher-Freeman-Halton exact test with regard to the distribution of participants' gender (p = 0.845), age (p = 0.460), familiarity with the building (p = 0.697) or previous VR experience (p = 0.549) over the four different scenarios.

5.2. Participant's perception of the virtual environment

Total scores were calculated for each questionnaire of each participant with respect to the Simulator Sickness Questionnaire (SSQ), System Usability Scale (SUS) and Presence Questionnaire (PQ), as shown in Table 4. In order to test significant differences of questionnaire results among four scenarios, the non-parametric Kruskal–Wallis was conducted for SSQ, and one-way ANOVA was used to compare the results of the SUS and the PQ between the participant groups. No significant difference was found in the SSQ among four scenarios, H (3) = 3.044, p = 0.385, as well as in the SUS, F (3, 90) = 2.452, p = 0.068 and in the PQ, F (3, 90) = 1.519, p = 0.215. This finding signals that the differences in perception between the four scenarios are very limited.

Furthermore, the mean scores and standard deviations of each questionnaire results were calculated (Table 5). The score of SSQ reflects the symptomatology of participants' experience in the virtual environment. The maximum total score of the SSQ is 236 (Kennedy et al., 2003). In general, the SSQ score of all participants was quite low in the present study. Table 6 shows the mean scores and standard deviations of each sub-scale in SSQ. It shows that the subscale of Nausea received the lowest score, increased by Oculomotor and Disorientation. Although the disorientation subscale is related to vestibular disturbances, such as dizziness and vertigo, it may also be an indicator of having experienced higher levels of virtual presence (Barfield and Weghorst, 1993). Meanwhile, none of the participants got sick during the experiment nor showed any symptoms, such as dizziness or nausea.

Besides that, the score of the Presence Questionnaire was obtained summing the responses of 22 items, scores can range from a minimum of

Table 3

Demogran	hic	informa	tion (∩f '	participants.
Demograph	inc .	mornio	unon (,	pui deipuito.

Descriptive	Category	Scenarios			
information		No information	Exit sign	Direction	Validation
Gender	Male	11	17	13	16
	Female	10	10	7	10
Age	<18	2	0	2	0
	18-25	9	15	10	16
	26-35	8	8	7	7
	36–45	0	2	1	3
	>45	2	2	0	0
Familiarity with the	Not at all familiar	6	11	11	12
building	Slightly familiar	10	11	5	10
	Moderately familiar	4	4	3	2
	Very familiar	1	0	0	0
	Extremely familiar	0	1	1	2
Previous	Never	9	7	5	9
Experience	Occasionally	9	13	13	13
with VR	Frequently	2	7	2	3
	Usually	0	0	0	1
	Always	1	0	0	0

Table 4

The mean score and standard deviations of each questionnaire results in four scenarios.

Scenarios		SSQ	PQ	SUS
No information	$\text{Mean}\pm\text{SD}$	$\textbf{33.84} \pm \textbf{37.52}$	$\textbf{72.71} \pm \textbf{12.76}$	62.74 ± 8.06
Exit sign	$\text{Mean} \pm \text{SD}$	$\textbf{25.07} \pm \textbf{25.72}$	$\textbf{73.07} \pm \textbf{12.77}$	$\textbf{70.00} \pm \textbf{13.03}$
Direction	$\text{Mean}\pm\text{SD}$	23.19 ± 25.81	65.95 ± 14.08	73.75 ± 12.34
Validation	$\text{Mean}\pm\text{SD}$	20.14 ± 25.17	$\textbf{72.81} \pm \textbf{12.16}$	$\textbf{71.06} \pm \textbf{18.35}$

Table 5

m1 1	. 1 1	1	c 1	
The mean scores and	standard	deviations of	t each	duestionnaire.

	SSQ	PQ	SUS
$\text{Mean}\pm\text{SD}$	25.26 ± 28.61	$\textbf{71.40} \pm \textbf{12.99}$	69.47 ± 14.07

Table 6

The mean and standard deviations of subscales in SSQ.

Subscale	Nausea	Oculomotor	Disorientation
$\text{Mean} \pm \text{SD}$	19.68 ± 34.25	25.31 ± 30.88	$\textbf{35.91} \pm \textbf{53.41}$

19 to a maximum of 133. The total scores of PQ with a mean of 71.40 and a standard deviation of 12.99 (Table 5) indicated that participants experienced a moderate amount of presence in the VR experiment. The mean scores and standard deviations of the five sub-factors in the PQ questionnaire (Witmer and Singer, 1998) were evaluated in more detail. As shown in Table 7, the interface quality receives the highest score, which indicates that the device had little distraction for participants performing their assignments. Meanwhile, the involvement factor also receives a relatively high score, which shows that participants experienced consistent concentration on the assignments they need to perform in the VR experiment. Last of all, the score of the System Usability Scale was calculated via multiplying the sum of the converted responses by 2.5. A SUS score above 68 identifies that the tool's usability is above average (Sauro, 2011). The mean score of SUS in the current experiment is 69.47, which indicates the participants were satisfied with the usability of the VR system.

5.3. Exit choice behaviour in four scenarios

The exit choice behaviour of the participants was assessed in three parts, namely the number of exits they identified after the VR experiment (i.e., the choice set), the actual exit that has been chosen by the participants (i.e., the exit choice), and the impact of information on both elements of the exit choice behaviour.

5.3.1. The number of identified exits

In the questionnaire, participants filled in the number of identified exits and drew these exits on a sheet of paper. If the location at which a participant drew an exit was reasonably close to the actual location of real-life exits, the exit was counted as one of the correctly identified exits. The results of the number of identified exits per scenario are depicted in Table 8, which shows that, on average, participants from the Exit sign scenario identified the most exits (M = 5.81, SD = 2.11). As expected, the average number of exits identified by participants in the

ble	7					
-----	---	--	--	--	--	--

Breakdown of the results of the Presence Questionnaire (range from 1 to 7).

	Involvement	Immersion	Visual fidelity	Interface quality ^a	Sound effect
$\begin{array}{c} \text{Mean} \pm \\ \text{SD} \end{array}$	3.24 ± 0.86	$\begin{array}{c} 3.20 \ \pm \\ 0.72 \end{array}$	$\begin{array}{c} \textbf{2.88} \pm \\ \textbf{0.99} \end{array}$	$\textbf{4.61} \pm \textbf{1.00}$	$\begin{array}{c} \textbf{2.68} \pm \\ \textbf{1.41} \end{array}$

^a Reversed items.

Та

Y. Feng et al.

Table 8

The number of identified exits.

$an \pm SD$ Ma	iximum N	Minimum		
2 ± 1.12 5	1			
1 ± 2.11 8	1			
0 ± 1.47 8	2			
8 ± 1.24 6	1			
	an \pm SD Ma 2 ± 1.12 5 1 ± 2.11 8 0 ± 1.47 8 8 ± 1.24 6	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		

scenario with direction information was higher than the number of exits identified in the No information scenario. At the same time, in the No information scenario, participants identified the least exits (M = 2.62, SD = 1.12).

The Kolmogorov–Smirnov test revealed that the number of identified exits in the Exit sign scenario and the Direction scenario did not significantly deviate from a normal distribution, but in the No information scenario and the Validation scenario was non-normally distributed. Thus the impact of information on the number of recognised exits for each scenario was analysed using the non-parametric Kruskal–Wallis test and in combination with Dunn post hoc test.

Results from Kruskal-Wallis test demonstrated that there was a significant difference of the mean number of identified exits among four scenarios (H (3) = 22.895, p < 0.001), which provided strong evidence of a difference between the mean number of identified exits at least one pair of scenarios. Afterwards, Dunn's post hoc test was conducted to test pairwise comparisons, namely the between scenario differences for the number of identified exits. The results showed that there are significant differences of the number of identified exits between the Exit sign scenario and the Validation scenario (p < 0.001), the Exit sign scenario and the Direction scenario (p = 0.001), and the Exit sign scenario and the No information scenario (p < 0.001). Meanwhile, there was no evidence of a significant difference between other pair scenarios. This finding identifies that participants tend to recognise and identify more exits when additional information about exit sign was provided than if there was no information, directional information or the presence of other pedestrians. This finding confirms that increasing the visibility of exits has a significant influence on pedestrian identifying the exits during evacuations.

5.3.2. Final exit choice

On the same paper where the participants drew the location of exits, they also indicated which exit was their final exit choice, which is presented in Table 9. Even though eight exits were available, the exit choices were mainly distributed over exits A1, A2, C and D, which are the nearest exits compared to exit E, F and G. Besides that, the results show that participants had a variety of final exit choices amongst the four scenarios. As one can see, in these experimental scenarios (i.e., No information, Exit sign, Direction) a wider variation of final exit choices was considered than in the Validation scenario (i.e., exits A1 and C were most chosen). At the same time, when there was no additional information provided (i.e., the No information scenario) exits A1 and D were most often chosen. In the Direction scenario, the majority of participants chose exits C and D. Yet, in the increased visibility of exits in the Exit sign scenario, there is a fairly even division in terms of exit choices. Compared to the other three scenarios, the distribution of the participant's final exit choice in the Exit sign scenario was more symmetrical.

In order to test whether different types of information significantly influence participants' exit choice, a the Fisher-Freeman-Halton exact test was conducted. The results revealed that participants' final exit choice in the Validation scenario is significantly different to the No information scenario (p < 0.001), the Exit sign scenario (p = 0.002), and the Direction scenario (p = 0.001). These results suggested that among the four scenarios, the presence and choices of other pedestrians in the Validation scenario had a significant impact on participants' final exit choice. Meanwhile, there was a significant difference between the No information scenario and the Direction scenario (p = 0.013). However, there were no significant differences either between No information scenario and Exit sign scenario (p = 0.137), or Exit sign scenario and Direction scenarios (p = 0.266). This indicates that compared to the scenario without any additional information, directional information and other pedestrians both had significant influences on the participant's exit choice. At the same time, increased visibility of exits did not have a significant influence on final exit choice.

6. Discussions

6.1. Discussion pertaining to the validation results

The results and findings of the exit choice behaviour from the field experiment and the VR experiment were presented in section 4. These findings illustrate that the methodological differences between the two experiments do not result in significant differences regarding the pedestrians' exit choice behaviour.

Up to this point, only a few studies validated pedestrian evacuation behaviour collected by means of VR experiments, most of which used a projection VR (Kobes et al., 2010a) or an HMD with a simulated virtual environment (Kinateder and Warren, 2016). The current validation study is complementary to studies showing that results regarding pedestrian exit choice behaviour are valid through different types of VR equipment. By comparing exit choice of participants in a real-life evacuation drill and an identical virtual environment, preliminary steps are taken to validate this particular VR simulator (smartphonebased HMD and 360° video) to study pedestrian exit choice behaviour during evacuations. Please note, that this paper only provides the first step towards comprehensively establishing the validity of this VR simulator, namely ecological validity. More research is required to also establish the content, construct and criterion validity of this particular VR simulator. The current VR experiment still includes several confounding factors (e.g., level of familiarity and age distributions) and only includes one repetition of one particular scenario. It is important to, in follow-up research, attempt to eliminate these issues via a quasicontrolled research setup.

Moreover, the findings highlight the influence of herding behaviour on pedestrian exit choice, where pedestrians choose to follow others during evacuations. In agreement with previous work (Helbing et al., 2000; Lovreglio et al., 2016; Moussaïd et al., 2016), herding behaviour was found in both the field experiment and VR experiment. While we found other factors influenced pedestrian exit choice (e.g., distance to

Table 9
The participants' final exit choices for all four scenarios.

Exits	A1	A2	В	С	D	E	F	G	Total
No information	6	4	0	1	8	0	2	0	21
	(28.57%)	(19.05%)	(0%)	(4.76)	(38.10)	(0%)	(9.52)	(0%)	
Exit sign	5	5	0	5	5	5	1	1	27
	(18.52%)	(18.52%)	(0%)	(18.52%)	(18.52%)	(18.52%)	(3.70%)	(3.70%)	
Direction	3	0	0	5	9	3	0	0	20
	(15.00%)	(0%)	(0%)	(25.00%)	(45.00%)	(15.00%)	(0%)	(0%)	
Validation	13	0	0	10	0	3	0	0	26
	(50.00%)	(0%)	(0%)	(38.46%)	(0%)	(11.54%)	(0%)	(0%)	

exits, visibility of exits), herding behaviour was a main influential factor consistent across in both experiments, in spite of the fact that the individuals from the field experiment were very familiar with the environment, while the participants in the VR experiment had a relatively low familiarity with the building.

6.2. Discussion pertaining to the effect of information

Section 5 presented the findings regarding the effect of different types of information on pedestrian exit choice behaviour amongst four different scenarios in the VR experiment, as well as the results regarding participants perception of the virtual environment. Now we turn to the implications of these findings for deepening our understanding of pedestrian exit choice behaviour during evacuations and application of VR experiments for pedestrian safety research.

First, this study shows that pedestrians were more likely to recognise exits during evacuations when additional information was provided in the environment. Especially increasing the visibility of the exit sign allows pedestrians to, on average, recognise more exits. These results are consistent with studies demonstrating the effect of the visibility of exit signs on pedestrian's recognition of exits during evacuations (Kobes et al., 2010a; Tang et al., 2009; Wong and Lo, 2007).

Secondly, the results suggest that the exits are not evenly used. This result is in line with other studies that look at the influence of spatial distribution of exits on pedestrian exit choice (Haghani and Sarvi, 2016a; Liao et al., 2014). Similarly, when other people are present, the usage of the exits is also asymmetrical. Moreover, increased visibility of the exits is also found to ensure a more symmetrical usage of the exits. Besides that, as suggested in the literature, this study finds that pedestrians were overall more likely to choose the nearest exits (Feng et al., 2020; Guo et al., 2012; Haghani and Sarvi, 2016b; Kobes et al., 2010a; Li et al., 2019; Liao et al., 2017).

Third, this study shows that although good visibility of exit sign helps participants identify exits in the environment, it does not have a significant influence on their final exit choice. Instead, other pedestrians and directional information have significant influences on the participant's final exit choice. That is, participants were more likely to choose the exit which other pedestrians had already chosen. They perceived other individuals as potential sources of information. These findings are in line with the results of studies that researched the impact of other pedestrians on pedestrians' exit choice (Kinateder et al., 2018; Kinateder and Warren, 2016). Moreover, this study also finds that pedestrians prefer to choose the exit where the directional signs point to. These findings are complementary to studies that only investigated one type of information strategy on pedestrian exit choice behaviour during evacuations (e.g., Wong and Lo, 2007).

Finally, the results demonstrate that the current VR experiment was applicable to study pedestrian exit choice behaviours during evacuations. Primarily, the ecological validity is established via comparing pedestrian exit choice in real-life evacuation drill and VR experiment. Moreover, the questionnaire results illustrated that the virtual environment was relatively immersive and the equipment's usability was sufficient for the purpose of this experiment. It also reports low simulator sickness. These results show that this experimental setup can be used as a research tool to study exit behaviour during evacuations. Yet, different from more expensive or complex VR systems, our VR experiment used a combination of HMD device and a 360° video-recording of the real-life evacuation drills. The fact that this rather 'simple' experimental setup provides valid results suggests that the choice of suitable VR devices may depend more on the behavioural data researchers want to collect than the sophistication of the VR equipment.

7. Conclusion and future work

This paper investigates pedestrian exit choice behaviour during evacuations using both VR experiments and field experiment. In particular, four different information strategies were realised in the VR experiment and the effects of these strategies on pedestrian exit choice behaviour were examined. This included three types of information, namely increased visibility of exit signs, directional signs to outside, and information provided by other pedestrians.

Firstly, this study provides preliminary proof of the ecological validity of the VR simulator using the results from a field experiment and a similar VR scenario. The results demonstrated that the combination of smartphone-based HMD and 360° video can be used to conduct evacuation experiments under controlled experimental conditions, which allows researchers to control specific variables of interest systematically and to test pedestrian exit choice behaviour in well-specified scenarios (Kinateder et al., 2018). Secondly, the impact of different types of information strategies was investigated using the validated VR experiment method. It was found that the presence of other people and information pertaining to the direction of the exits has a significant influence on pedestrian exit choice. Moreover, pedestrians' exit choice was found to be asymmetrical, especially in the scenario involving other pedestrians. Furthermore, the behavioural findings of this research and questionnaire data pertaining to the usability of this particular VR system add new much-needed insights to the discussion around the validation and usage of VR for pedestrian behaviour research.

These results provide useful information to evacuation management officials to develop (better) strategies to facilitate efficient evacuation (e.g., how to use exits evenly to avoid crowd in front of exits) in an environment with an open field and multiple exits. Moreover, this study provides a new VR tool that can be used to test new information strategies before they are implemented in the field in order to determine their effectiveness and/or train evacuation management personnel.

There are several limitations in this study. Firstly, even though we have attempted to recreate the scenario as close as possible, differences exist between the real-life evacuation drill and the validation scenarios in the VR experiments. That is, in real life, pedestrians were able to navigate freely and interact with other people, while the participants in the VR experiment only have visual rotation within the captured 360° field of view. Besides that, the distance between the exits and pedestrians is different in real life (i.e., next to the tables) than in the virtual environment (i.e., the middle of the room). Moreover, although limited influences of the environmental factors were expected, they could also not be completely excluded in the festival setting. In future studies, it is better to choose a more isolated environment to separate participants from the real-life environment, such as noise, lighting, sound. Secondly, although the used mobile-based HMD device was easier to operate and quicker to set up than more elaborate VR devices, some features, such as eye-tracking and free movements, are not available in this 'simple' VR device. This means that some aspects of pedestrian evacuation behaviour could not be recorded nor analysed in this study (e.g., preevacuation behaviour, gazing behaviour). Thirdly, the questionnaire data revealed that participants experienced a relatively lower feeling of presence compared to more advanced VR devices (e.g., Deb et al., 2017). More research (e.g., experiments with similar scenario settings) is required to establish whether the static vantage point and the limited set of features of this VR simulator do not hamper other types of validity (e. g., construct, discriminant and internal validity).

Yet, even when accounting for these limitations, there is benefit in using a low-cost HMD device, VR environment setup and relatively comparable conditions for pedestrian safety research. Especially, when unclarity exists regarding which independent variables should be included in the comprehensive pedestrian behaviour study, a preliminary pilot study featuring the combination of 360° video and smartphone-based HMD can allow researchers to get more grip on the actual set of influential factors. However, the last limitation of this VR experiment setup (i.e., lower level of presence) is difficult to solve using the current device. Thus, the next steps in this research featuring exit choice behaviour will be to enhance participants' feeling of presence in the virtual environment by using a more sophisticated smartphonebased VR device and involving interactions with the virtual environment.

Declaration of Competing Interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Acknowledgement

We thank the China Scholarship Council for their financial contribution. Meanwhile, this research is also supported by the ALLEGRO project, which is financed by the European Research Council (Grant Agreement No. 669792).

References

- Barfield, W., Weghorst, S., 1993. The Sense of Presence within Virtual Environments: A Conceptual Framework. Adv. Human Factors Ergon. 19, 699–699.
- Bode, N.W.F., Kemloh Wagoum, A.U., Codling, E.A., 2015. Information use by humans during dynamic route choice in virtual crowd evacuations. R. Soc. Open Sci. 2 https://doi.org/10.1098/rsos.140410.
- Bode, N.W.F., Wagoum, A.U.K., Codling, E.A., 2014. Human responses to multiple sources of directional information in virtual crowd evacuations. J. R. Soc. Interface 11. https://doi.org/10.1098/rsif.2013.0904.
- Brooke, J., 1996. SUS A quick and dirty usability scale. Usability Eval. Ind. 189, 4–7. https://doi.org/10.1002/hbm.20701.
- Cao, L., Lin, J., Li, N., 2019. A virtual reality based study of indoor fire evacuation after active or passive spatial exploration. Comput. Human Behav. 90, 37–45. https://doi. org/10.1016/j.chb.2018.08.041.
- Cao, S., Fu, L., Wang, P., Zeng, G., Song, W., 2018. Experimental and modeling study on evacuation under good and limited visibility in a supermarket. Fire Saf. J. 102, 27–36. https://doi.org/10.1016/j.firesaf.2018.10.003.
- Cosma, G., Ronchi, E., Nilsson, D., 2016. Way-finding lighting systems for rail tunnel evacuation: A virtual reality experiment with Oculus Rift®. J. Transp. Saf. Secur. 8, 101–117. https://doi.org/10.1080/19439962.2015.1046621.
- Deb, S., Carruth, D.W., Sween, R., Strawderman, L., Garrison, T.M., 2017. Efficacy of virtual reality in pedestrian safety research. Appl. Ergon. 65, 449–460. https://doi. org/10.1016/j.apergo.2017.03.007.
- Duarte, E., Rebelo, F., Teles, J., Wogalter, M.S., 2014. Behavioral compliance for dynamic versus static signs in an immersive virtual environment. Appl. Ergon. 45, 1367–1375. https://doi.org/10.1016/j.apergo.2013.10.004.
- Duives, D., Mahmassani, H., 2012. Exit Choice Decisions During Pedestrian Evacuations of Buildings. Transp. Res. Rec. J. Transp. Res. Board 2316, 84–94. https://doi.org/ 10.3141/2316-10.
- Fang, Z., Gao, H., Jiang, L., Li, X., Lv, W., 2020. Human movement characteristics during emergency evacuations in a virtual environment. Fire Saf. J. 115, 103147. https:// doi.org/10.1016/j.firesaf.2020.103147.
- Fang, Z., Song, W., Zhang, J., Wu, H., 2010. Experiment and modeling of exit-selecting behaviors during a building evacuation. Physica A 389, 815–824. https://doi.org/ 10.1016/j.physa.2009.10.019.
- Feng, Y., Duives, D., Daamen, W., Hoogendoorn, S., 2021. Data collection methods for studying pedestrian behaviour: A systematic review. Build. Environ. 187, 107329. https://doi.org/10.1016/j.buildenv.2020.107329.
- Feng, Y., Duives, D.C., Hoogendoorn, S.P., 2020. Developing a VR tool for studying pedestrian movement and choice behavior. In: 2020 IEEE Conference on Virtual Reality and 3D User Interfaces Abstracts and Workshops (VRW), pp. 814–815. https://doi.org/10.1109/VRW50115.2020.00258.
- Fridolf, K., Ronchi, E., Nilsson, D., Frantzich, H., 2013. Movement speed and exit choice in smoke-filled rail tunnels. Fire Saf. J. 59, 8–21. https://doi.org/10.1016/j. firesaf.2013.03.007.
- Galea, E.R., Deere, S.J., Hopkin, C.G., Xie, H., 2017a. Evacuation response behaviour of occupants in a large theatre during a live performance. Fire Mater. 41, 467–492. https://doi.org/10.1002/fam.2424.
- Galea, Edwin R., Xie, H., Deere, S., Cooney, D., Filippidis, L., 2017b. Evaluating the effectiveness of an improved active dynamic signage system using full scale evacuation trials. Fire Saf. J. 91, 908–917. https://doi.org/10.1016/j. firesaf.2017.03.022.
- Guo, R.Y., Huang, H.J., Wong, S.C., 2012. Route choice in pedestrian evacuation under conditions of good and zero visibility: Experimental and simulation results. Transp. Res. Part B Methodol. 46, 669–686, https://doi.org/10.1016/j.trb.2012.01.002.
- Haghani, M., Sarvi, M., 2016a. Pedestrian crowd tactical-level decision making during emergency evacuations. J. Adv. Transp. 50, 1870–1895. https://doi.org/10.1002/ atr.1434.

Haghani, M., Sarvi, M., 2016b. Human exit choice in crowded built environments: Investigating underlying behavioural differences between normal egress and emergency evacuations. Fire Saf. J. 85, 1–9. https://doi.org/10.1016/j. firesaf.2016.07.003.

Helbing, D., Farkas, I., Vicsek, T., 2000. Simulating dynamical features of escape panic. Nature 407, 487–490.

- Heliövaara, S., Kuusinen, J.M., Rinne, T., Korhonen, T., Ehtamo, H., 2012. Pedestrian behavior and exit selection in evacuation of a corridor - An experimental study. Saf. Sci. 50, 221–227. https://doi.org/10.1016/j.ssci.2011.08.020.
- Imanishi, M., Sano, T., 2019. Route choice and flow rate in theatre evacuation drills: Analysis of walking trajectory data-set. Fire Technol. 55, 569–593. https://doi.org/ 10.1007/s10694-018-0783-2.
- Kennedy, R.S., Drexler, J.M., Compton, D.E., Stanney, K.M., Lanham, D.S., Harm, D.L., 2003. Configural scoring of simulator sickness, cybersickness, and space adaptation syndrome: Similarities and differences. In: Virtual and Adaptive Environments: Applications, Implications, and Human Performance Issues. pp. 247–278. doi: 10 .1201/9781410608888.ch12.
- Kennedy, R.S., Lane, N.E., Berbaum, K.S., Lilienthal, M.G., 1993. Simulator sickness questionnaire: An enhanced method for quantifying simulator sickness. Int. J. Aviat. Psychol. 3, 203–220. https://doi.org/10.1207/s15327108ijap0303_3.
- Kinateder, M., Comunale, B., Warren, W.H., 2018. Exit choice in an emergency evacuation scenario is influenced by exit familiarity and neighbor behavior. Saf. Sci. 106, 170–175. https://doi.org/10.1016/j.ssci.2018.03.015.
- Kinateder, M., Müller, M., Jost, M., Mühlberger, A., Pauli, P., 2014a. Social influence in a virtual tunnel fire – Influence of conflicting information on evacuation behavior. Appl. Ergon. J. 45, 1649–1659. https://doi.org/10.1016/j.apergo.2014.05.014.
- Kinateder, M., Ronchi, E., Gromer, D., Müller, M., Jost, M., Nehfischer, M., Mühlberger, A., Pauli, P., 2014b. Social influence on route choice in a virtual reality tunnel fire. Transp. Res. Part F Traffic Psychol. Behav. 26, 116–125. https://doi.org/ 10.1016/j.trf.2014.06.003.
- Kinateder, M., Warren, W.H., 2016. Social influence on evacuation behavior in real and virtual environments. Front. Robot. AI 3, 1–8. https://doi.org/10.3389/ froht.2016.00043.
- Kinateder, M., Warren, W.H., Schloss, K.B., 2019. What color are emergency exit signs? Egress behavior differs from verbal report. Appl. Ergon. 75, 155–160. https://doi. org/10.1016/j.apergo.2018.08.010.
- Kobes, M., Helsloot, I., De Vries, B., Post, J., 2010a. Exit choice, (pre-)movement time and (pre-)evacuation behaviour in hotel fire evacuation - Behavioural analysis and validation of the use of serious gaming in experimental research. Procedia Eng. 3, 37–51. https://doi.org/10.1016/j.proeng.2010.07.006.
- Kobes, M., Helsloot, I., de Vries, B., Post, J.G., Oberijé, N., Groenewegen, K., 2010b. Way finding during fire evacuation; an analysis of unannounced fire drills in a hotel at night. Build. Environ. 45, 537–548. https://doi.org/10.1016/j. buildenv.2009.07.004.
- Krevelen, D.W.F. van, Poelman, R., 2010. A survey of augmented reality technologies, applications and limittions. Int. J. Virtual Real. 9, 1–20. https://doi.org/10.1155/ 2011/721827.
- Li, H., Zhang, J., Xia, L., Song, W., Bode, N.W.F., 2019. Comparing the route-choice behavior of pedestrians around obstacles in a virtual experiment and a field study. Transp. Res. Part C Emerg. Technol. 107, 120–136. https://doi.org/10.1016/j. trc.2019.08.012.
- Liao, W., Wagoum, A.U.K., Bode, N.W.F., 2017. Route choice in pedestrians: Determinants for initial choices and revising decisions. J. R. Soc. Interface 14. https://doi.org/10.1098/rsif.2016.0684.
- Liao, W., Zheng, X., Cheng, L., Zhao, Y., Cheng, Y., Wang, Y., 2014. Layout effects of multi-exit ticket-inspectors on pedestrian evacuation. Saf. Sci. 70, 1–8. https://doi. org/10.1016/j.ssci.2014.04.015.
- Lin, J., Cao, L., Li, N., 2019. Assessing the influence of repeated exposures and mental stress on human wayfinding performance in indoor environments using virtual reality technology. Adv. Eng. Informatics 39, 53–61. https://doi.org/10.1016/j. aei.2018.11.007.
- Lin, J., Zhu, R., Li, N., Becerik-Gerber, B., 2020. Do people follow the crowd in building emergency evacuation? A cross-cultural immersive virtual reality-based study. Adv. Eng. Informatics 43, 101040. https://doi.org/10.1016/j.aei.2020.101040.
- Lovreglio, R., Borri, D., Dell'Olio, L., Ibeas, A., 2014. A discrete choice model based on random utilities for exit choice in emergency evacuations. Saf. Sci. 62, 418–426. https://doi.org/10.1016/j.ssci.2013.10.004.
- Lovreglio, R., Fonzone, A., dell'Olio, L., Borri, D., 2016. A study of herding behaviour in exit choice during emergencies based on random utility theory. Saf. Sci. 82, 421–431. https://doi.org/10.1016/j.ssci.2015.10.015.
- Moussaïd, M., Kapadia, M., Thrash, T., Sumner, R.W., Gross, M., Helbing, D., Hölscher, C., Ho, C., 2016. Crowd behaviour during high-stress evacuations in an immersive virtual environment. J. R. Soc. Interface 13, 414. https://doi.org/ 10.1098/rsif.2016.0414.
- Mujber, T.S., Szecsi, T., Hashmi, M.S.J., 2004. Virtual reality applications in manufacturing process simulation. J. Mater. Process. Technol. 155, 1834–1838. https://doi.org/10.1016/j.jmatprotec.2004.04.401.
- Proulx, G., 1995. Evacuation time and movement in apartment buildings. Fire Saf. J. 24, 229–246. https://doi.org/10.1016/0379-7112(95)00023-M.
- Rahouti, A., Lovreglio, R., Gwynne, S., Jackson, P., Datoussaïd, S., Hunt, A., 2020. Human behaviour during a healthcare facility evacuation drills: Investigation of preevacuation and travel phases. Saf. Sci. 129, 104754. https://doi.org/10.1016/j. ssci.2020.104754.
- Ronchi, E., Fridolf, K., Frantzich, H., Nilsson, D., Walter, A.L., Modig, H., 2018. A tunnel evacuation experiment on movement speed and exit choice in smoke. Fire Saf. J. 97, 126–136. https://doi.org/10.1016/j.firesaf.2017.06.002.
- Ronchi, E., Mayorga, D., Lovreglio, R., Wahlqvist, J., Nilsson, D., 2019. Mobile-powered head-mounted displays versus cave automatic virtual environment experiments for evacuation research. Comput. Animat. Virtual Worlds. https://doi.org/10.1002/ cav.1873.
- Ronchi, E., Nilsson, D., Kojić, S., Eriksson, J., Lovreglio, R., Modig, H., Walter, A.L., 2016. A virtual reality experiment on flashing lights at emergency exit portals for road

Y. Feng et al.

tunnel evacuation. Fire Technol. 52, 623–647. https://doi.org/10.1007/s10694-015-0462-5.

Sauro, J., 2011. SUStisfied? little-known system usability scale facts. UX 10 (3), 2011–2013.

- Shields, T.J., Boyce, K.E., 2000. A study of evacuation from large retail stores. Fire Saf. J. 35, 25–49. https://doi.org/10.1016/S0379-7112(00)00013-8.
- Silva, J.F., Almeida, J.E., Rossetti, R.J.F., Coelho, A.L., 2013. A serious game for EVAcuation training. In: IEEE 2nd International Conference on Serious Games and Applications for Health (SeGAH), pp. 1–6. https://doi.org/10.1109/ SeGAH.2013.6665302.
- Tang, C.-H., Wu, W.-T., Lin, C.-Y., 2009. Using virtual reality to determine how emergency signs facilitate way-finding. Appl. Ergon. 40, 722–730. https://doi.org/ 10.1016/j.apergo.2008.06.009.
- Tucker, A., Marsh, K.L., Gifford, T., Lu, X., Luh, P.B., Astur, R.S., 2018. The effects of information and hazard on evacuee behavior in virtual reality 99, 1–11. https://doi. org/10.1016/j.firesaf.2018.04.011.
- Vilar, E., Rebelo, F., Noriega, P., Teixeira, L., Duarte, E., Filgueiras, E., 2013. Are emergency egress signs strong enough to overlap the influence of the environmental

variables?. In: International Conference of Design, User Experience, and Usability, pp. 205–214. https://doi.org/10.1007/978-3-642-39238-2_23.

- Witmer, B.G., Singer, M.J., 1998. Measuring presence in virtual environments: A Presence Questionnaire. Presence Teleoperators Virtual Environ. 7, 225–240. https://doi.org/10.1145/985921.985934.
- Wong, L.T., Lo, K.C., 2007. Experimental study on visibility of exit signs in buildings. Build. Environ. 42, 1836–1842. https://doi.org/10.1016/j.buildenv.2006.02.011.
- Zhao, H., Thrash, T., Kapadia, M., Wolff, K., Holscher, C., Helbing, D., Schinazi, V.R., 2020. Assessing crowd management strategies for the 2010 Love Parade disaster using computer simulations and virtual reality. J. R. Soc. Interface 17. https://doi. org/10.1098/rsif.2020.0116.
- Zhu, K.J., Shi, Q., 2016. Experimental study on choice behavior of pedestrians during building evacuation. Procedia Eng. 135, 206–215. https://doi.org/10.1016/j. proeng.2016.01.110.
- Zhu, R., Lin, J., Becerik-Gerber, B., Li, N., 2020. Influence of architectural visual access on emergency wayfinding: A cross-cultural study in China, United Kingdom and United States. Fire Saf. J. 113, 102963. https://doi.org/10.1016/j. firesaf.2020.102963.